Author's response to reviews

Title: Cost per case or total cost? The potential of prevention of hand injuries in young children - retrospective and prospective studies

Authors:

Elinor Ljungberg (Elinor.Ljungberg@med.lu.se)
Katarina Steen Carlsson (Katarina.Steen_Carlsson@med.lu.se)
Lars B Dahlin (Lars.Dahlin@med.lu.se)

Version: 4 Date: 18 June 2008

Author's response to reviews:

Dear Editor,

Thank you for sending the reviewer's comments of our revised manuscript. We very much appreciate the very kind and positive letter from the new reviewer. It is very unusual that you get such an encouraging response from a reviewer. We are delighted to hear that the manuscript is acceptable for publication.

We have now included the requested information regarding ethical permission based on the approval from the Ethical Committee at Lund University in Sweden including the two project numbers.

If there are anything else we have to provide or change please contact via e-mail.

Sincerely yours

Lars Dahlin